Author's response to reviews

Title: Effectiveness of internet-supported cognitive behavioral and chronobiological interventions and effect-moderation by insomnia subtype: study protocol of a randomized controlled trial

Authors:

   Kim Dekker (k.dekker@nin.knaw.nl)
   Jeroen S Benjamins (j.s.benjamins@uu.nl)
   Annemieke Van Straten (a.van.straten@vu.nl)
   Winni F Hofman (w.f.hofman@uva.nl)
   Eus JW Van Someren (e.van.someren@nin.knaw.nl)

Version: 2
Date: 1 June 2015

Author's response to reviews:

Dear editor,

please find my revised manuscript. Revisions are the format of the title and the added date of trial registration in the abstract.

On behalf of all co-authors,

Kim Dekker